Author’s response to reviews

Title: Rationale and Design of Extended Cancer Education for Longer Term Survivors (EXCELS): A Randomized Control Trial of ‘High Touch’ vs. ‘High Tech’ Cancer Survivorship Self-management Tools in Primary Care

Authors:
Stacy Davis (sdavis1908@yahoo.com; stacy.davis@rutgers.edu)
Denalee O’Malley (omalledm@rwjms.rutgers.edu)
Alicja Bator (alicjab@rwjms.rutgers.edu)
Pamela Ohman-Strickland (ohmanpa@sph.rutgers.edu)
Lynn Clemow (clemowlp@rwjms.rutgers.edu)
Jeanne Ferrante (ferranjm@rwjms.rutgers.edu)
Benjamin Crabtree (crabtrbf@rwjms.rutgers.edu)
Suzanne Miller (suzanne.miller@fccc.edu)
Patricia Findley (pfindley@ssw.rutgers.edu)
Shawna Hudson (hudsonsh@rwjms.rutgers.edu)

Version: 2 Date: 25 Mar 2019

Author’s response to reviews:

March 21, 2019

Catherine Rice, PhD
Senior Assistant Editor
BMC Cancer

Dear Dr. Rice,

We are pleased to revise and resubmit our manuscript, Manuscript ID: BCAN-D-19-00195R1, titled “Rationale and Design of Extended Cancer Education for Longer Term Survivors (EXCELS): A Randomized Control Trial of ‘High Touch’ vs. ‘High Tech’ Cancer Survivorship Self-management Tools in Primary Care.”
BMC Cancer requested the following formatting changes, which have been addressed below and in the revised manuscript.

1. All files that we do not wish to see published have been removed from the file inventory.

2. The trial was not registered retrospectively and the registration number is NCT03233555.

3. Declarations: Ethics approval and consent to participate has been revised to include the full names of the ethics committees that approved the study and appropriate reference numbers.

4. Final manuscript has been uploaded as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethrough or text in different colors.

We look forward to hearing from you about the suitability of this revised manuscript for publication in BMC Cancer.

Kind regards,

Shawna V. Hudson, PhD

Professor and Research Division Chief, Department of Family Medicine and Community Health Director, Center Advancing Research and Evaluation for Patient-Centered Care (CARE-PC) Rutgers Robert Wood Johnson Medical School Rutgers, The State University of New Jersey 112 Paterson Street, Room 446 New Brunswick, NJ 08901